Author's response to reviews

Title: Designing effective visualizations of habits data to aid clinical decision making

Authors:

   Joost de Folter (joost.defolter@brunel.ac.uk)
   Hulya Gokalp (hulya.gokalp@brunel.ac.uk)
   Jo Fursse (j.fursse@gmail.com)
   Urvashi Sharma (urvashi.sharma@brunel.ac.uk)
   Malcolm Clarke (malcolm.clarke@brunel.ac.uk)

Version: 3  Date: 19 May 2014

Author's response to reviews: see over
Dear Sir/Madam,

Manuscript title: ‘Designing effective visualizations of habits data to aid clinical decision making’

Please accept our revised submission to the BMC journal of medical Informatics & Decision Making. We have added appropriate lines regarding ethics at the end of the methods section (section 2.4; page 5). Line and page numbers have been added as well. Finally we have corrected the Word references (or links) that had occurred while reformatting to the required format for submission.

Please do not hesitate to contact us if you require further information or documentation.

Yours faithfully,

Malcolm Clarke, Reader, Department of Computer Science, Brunel University